Author's response to reviews

Title: Metabolic phenotype of methylmalonic acidemia in mice and humans: the role of skeletal muscle

Authors:

Randy J Chandler (rchandler@mail.nih.gov)
Jennifer Sloan (jsloan@mail.nih.gov)
Hong Fu (kaestner@mail.med.upenn.edu)
Matthew Tsai (matt.tsai@gmail.com)
Sally Stabler (sally.Stabler@UCHSC.edu)
Robert Allen (robert.allen@uchsc.edu)
Klaus H Kaestner (kaestner@mail.med.upenn.edu)
Haig H Kazazian (kazazian@mail.med.upenn.edu)
Charles P Venditti (venditti@mail.nih.gov)

Version: 5 Date: 10 October 2007

Author's response to reviews: see over
October 10, 2007

Dear Members of the Editorial Board,

Our manuscript has been revised and reformatted per the suggestions of the editorial production team and is ready for publication.

Sincerely,

Charles P. Venditti MD, PhD
Sincerely yours,

Charles P. Venditti MD, PhD
Genetic Disease Research Branch
National Human Genome Research Institute
National Institutes of Health
Bldg 49, Room 4A56A
Bethesda, MD 20892-4472